Author's response to reviews

Title: Pregnancy in multiple system atrophy: a case report

Authors:

Lirong Zhu (zhul@neuro.wustl.edu)
Nigel J. Cairns (cairns@wustl.edu)
Samer D. Tabbal (tabbals@neuro.wustl.edu)
Brad A. Racette (racetteb@neuro.wustl.edu)

Version: 3 Date: 22 October 2011

Author's response to reviews: see over
October 22, 2011

The Journal of Medical Case Reports Editorial Team

Manuscript number: 1333955775694612
Corresponding author: Racette
Title: Pregnancy in multiple system atrophy: a case report

Dear Editorial Team,

We thank the reviewers for their critical comments and suggestions. We have edited the manuscript and made the requested revisions. We respond to the reviewers’ comments in point from below:

Reviewer 3:

1. The use of “alpha (greek symbol)” (first sentence of the abstract) versus “alpha” (elsewhere in the abstract, text and figure legend): You may want to use one or the other, depending on the Editorial Team’s suggestions.

_We have changed the “alpha (greek symbol)” in the abstract and first paragraph of the introduction to “alpha”._

2. Case Presentation, sixth paragraph: “All these symptoms raise the possibility of MSA-P”: Should this be written in the past: “All these symptoms raisED the possibility of MSA-P”? This is just a question.

_We have revised our manuscript accordingly._

3. Acknowledgments: “staff of the Departments of Neurology, the Department of Pathology…”: You may want to write: “staff of the Department (NO PLURAL) of Neurology AND the Department of Pathology…”

_We have corrected our manuscript accordingly._

4. Reference: ReferenceS (plural)?

_We have revised our manuscript accordingly._

In addition, the Editorial Team would also like to request for formatting changes required for your manuscript, as it does not conform to the journal style. Kindly address the following:

1. Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report.
We have changed the title to “Pregnancy in multiple system atrophy: a case report”

We appreciate the opportunity to revise this manuscript and hope that it is now acceptable for publication in The Journal of Medical Case Reports.

Sincerely,

Lirong Zhu, M.D., Ph.D.
Nigel J. Cairns, Ph.D.
Samer D. Tabbal, M.D.
Brad A. Racette, M.D.